Author's response to reviews

Title: Comparison of efficacy between incretin-based therapies for type 2 diabetes mellitus

Authors:

Kaustubh Nisal (knisal@gmail.com)
Ram Kela (drramkela@doctors.org.uk)
Kamlesh Khunti (kk22@le.ac.uk)
Melanie Davies (melanie.davies@uhl-tr.nhs.uk)

Version: 4 Date: 23 September 2012

Author's response to reviews:

Ursula D'Souza, PhD
The Senior Editor
BMC Medicine

Subject: Revision of Abstract of manuscript –September

Dear Editor,
Authors have revised the abstract as per the BMC Medicine guidelines.
Once again many thanks for the review. Please contact me should there be any problems.

Yours sincerely

Kaustubh Nisal
University Hospitals of Leicester
Leicester Royal Infirmary
Leicester LE1 5WW
knisal@gmail.com